Author's response to reviews

Title: Teng-Long-Bu-Zhong-Tang, a Chinese herbal formula, enhances anticancer effects of 5- Fluorouracil in CT26 colon carcinoma

Authors:

Shan Deng (ziziryoma@yahoo.com.cn)
Bing Hu (beearhu@hotmail.com)
Hong-Mei An (soatp@yahoo.com.cn)
Qin Du (lsduqin123@126.com)
Ling Xu (xulg67@yahoo.com.cn)
Ke-Ping Shen (shen0313@163.com)
Xiu-Feng Shi (shi_yazi@sina.com)
Meng-Meng Wei (xiaomifeng.wmm@hotmail.com)
Yang Wu (stillwy@163.com)

Version: 2 Date: 29 January 2013

Author's response to reviews:

Dear Editors:

In this version, we have made the following changes:
1. In Figure 4A, we have changed the fonts of Arial to Arial Narrow.
2. In Figure 6B, we have changed the colour.
3. In Figure 7A, we have changed the fonts of Arial Narrow to Arial.

Thanks very much for your attention to our manuscript.

Sincerely yours,

Bing Hu, MD, PhD.
Department of Oncology
Longhua Hospital
Shanghai University of Traditional Chinese Medicine
725 South Wanping Road
Shanghai 200032, P. R. China
Tel: 86-21-64385700
Fax: 86-21-64398310
E-mail: beearhu@hotmail.com